Author’s response to reviews

Title: Ethical challenges experienced by UK military medical personnel deployed to Sierra Leone (Operation GRITROCK) during the 2014-2015 Ebola outbreak: a qualitative study

Authors:

Heather Draper (h.draper@warwick.ac.uk)

Simon Jenkins (S.Jenkins.4@warwick.ac.uk)

Version: 2 Date: 26 Nov 2017

Author’s response to reviews:

Our responses to the editorial comments are made beneath each comment in CAPS

1 According to submission guideline for research article, the abstract must include the following separate sections:

• Background: the context and purpose of the study
• Methods: how the study was performed and statistical tests used
• Results: the main findings
• Conclusions: brief summary and potential implications

Please structure the Abstract accordingly

THIS HAS NOW BEEN DONE

2 Please rename the section heading 'Introduction' to 'Background' in the main text

THIS HAS NOW BEEN DONE
3 List of Abbreviations

As abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

THIS HAS NOW BEEN DONE

4 In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript

THIS HAS NOW BEEN DONE

5 Please ensure that both authors have read and approved the final version and add a statement confirming this in Authors’ Contributions

AS WE HAD DONE SO PREVIOUSLY WE ASSUMED THAT YOU WERE REQUESTING AN EXPANDED STATEMENT. WE HAVE EXPANDED THE STATEMENT

6 Upon resubmission, Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

NO TRACK CHANGES HAVE BEEN INCLUDED